Author's response to reviews

Title: A Thyroid Tumor extending to the Parapharyngeal Space

Authors:

Fikret Cetik (fikretcetik@yahoo.com.tr)
Demet Yazici (demetyazici@yahoo.com)
Aysun Uguz (aysunuz@yahoo.com)

Version: 4 Date: 15 February 2006

Author's response to reviews:

We have checked the article for grammatical errors and misspellings. We have changed the sections of the article as you have mentioned in your e-mail. The abstract of the article has been divided in three sections, introduction part has been named as background, discussion part has been named as conclusions and changes have been made in the references part.